Shockwave lithotripsy compared with ureteroscopic stone treatment for adults with ureteric stones: the TISU non-inferiority RCT

Ranan Dasgupta,¹ Sarah Cameron,² Lorna Aucott,³ Graeme MacLennan,² Mary M Kilonzo,⁴ Thomas BL Lam,^{5,6} Ruth Thomas,² John Norrie,⁷ Alison McDonald,² Ken Anson,⁸ James N'Dow,⁵ Neil Burgess,⁹ Charles T Clark,¹⁰ Francis X Keeley,¹¹ Sara J MacLennan,⁶ Kath Starr² and Samuel McClinton^{5*}

¹Department of Urology, Imperial College Healthcare NHS Trust, London, UK ²Centre for Healthcare Randomised Trials, University of Aberdeen, Aberdeen, UK ³Health Services Research Unit, University of Aberdeen, Aberdeen, UK ⁴Health Economics Research Unit, University of Aberdeen, Aberdeen, UK ⁵NHS Grampian, Department of Urology, Aberdeen Royal Infirmary, Aberdeen, UK ⁶Academic Urology Unit, University of Aberdeen, Aberdeen, UK ⁷Usher Institute of Population Health Sciences and Informatics, University of Edinburgh, Edinburgh, UK

- ⁸Department of Urology, St George's University Hospitals NHS Foundation Trust, London, UK
- ⁹Department of Urology, Norfolk and Norwich University Hospitals NHS Foundation Trust, Norwich, UK
- ¹⁰Stone Patient Advisory Group, Section of Endourology, British Association of Urological Surgeons, London, UK
- ¹¹Bristol Urological Institute, North Bristol NHS Trust, Bristol, UK

*Corresponding author samuel.mcclinton@nhs.scot

Declared competing interests of the authors: Lorna Aucott is a member of the Public Health Research Research Funding Board (2017–21). John Norrie reports grants from the University of Aberdeen and the University of Edinburgh during the conduct of the study, and declares membership of the following National Institute for Health Research (NIHR) boards: Cardiopulmonary Resuscitation Decision-Making Committee; Health Technology Assessment (HTA) Commissioning Board; HTA Commissioning Sub-Board (Expression of Interest); HTA Funding Boards Policy Group; HTA Post-Board Funding Teleconference; NIHR Clinical Trials Unit Standing Advisory Committee; NIHR HTA and Efficacy and Mechanism Evaluation (EME) Editorial Board; and the Pre-exposure Prophylaxis Impact Review Panel. Ken Anson reports grants from a NIHR HTA-funded study for SUSPEND (Spontaneous Urinary Stone Passage ENabled by Drugs) and the Percutaneous nephrolithotomy, flexible Ureterorenoscopy and Extracorporeal shockwave lithotripsy for lower pole kidney stones (PUrE) randomised controlled trial, he was co-applicant on both and principal investigator at the site. James N'Dow was a member of the HTA General Board (2016–March 2019). Francis X Keeley is a member of the Boston Scientific (Marlborough, MA, USA) Advisory Committee and is assisting with the development of a new ureteroscope (Olympus Medical, Hamburg, Germany). Kath Starr reports grants from the NIHR HTA programme during the conduct of the study.

Published March 2022 DOI: 10.3310/WUZW9042

Plain English summary

The TISU trial Health Technology Assessment 2022; Vol. 26: No. 19 DOI: 10.3310/WUZW9042

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

A pproximately 1 in 20 people suffers from kidney stones that pass down the urine drainage tube (ureter) into the urinary bladder and cause episodes of severe pain (ureteric colic). People with ureteric colic attend hospital for pain relief and diagnosis. Although most stones smaller than 10 mm eventually reach the bladder and are passed during urination, some get stuck and have to be removed using telescopic surgery (called ureteroscopic stone treatment) or shockwave therapy (called shockwave lithotripsy).

Ureteroscopic stone treatment involves passing a telescope-containing instrument through the bladder and into the ureter to fragment and/or remove the stone. This is usually carried out under general anaesthetic as a day case. For shockwave lithotripsy, the patient lies flat on a couch and the apparatus underneath them generates shockwaves that pass through the skin to the ureter and break the stones into smaller fragments, which can be passed naturally in the urine. This involves using X-ray or ultrasound to locate the stone, but can be carried out on an outpatient basis and without general anaesthetic. Telescopic surgery is known to be more successful at removing stones after just one treatment, but it requires more time in hospital and has a higher risk of complications than shockwave lithotripsy (however, shockwave lithotripsy may require more than one session of treatment).

Our study, the Therapeutic Interventions for Stones of the Ureter trial, was designed to establish if treatment for ureteric colic should start with telescopic surgery or shockwave therapy. Over 600 NHS patients took part and they were split into two groups. Each patient had an equal chance of their treatment starting with either telescopic surgery or shockwave lithotripsy, which was decided by a computer program (via random allocation). We counted how many patients in each group had further procedures to remove their stone. We found that telescopic surgery was 11% more effective overall, with an associated slightly better quality of life (10 more healthy days over the 6-month period), but was more expensive in an NHS setting. The finding of a lack of any significant additional clinical benefit leads to the conclusion that the more cost-effective treatment pathway is shockwave lithotripsy with telescopic surgery used only in those patients in whom shockwave lithotripsy is unsuccessful.

© Queen's Printer and Controller of HMSO 2022. This work was produced by Dasgupta *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.014

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, the Cochrane Library and Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 10/137/01. The contractual start date was in March 2013. The draft report began editorial review in March 2019 and was accepted for publication in January 2020. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2022. This work was produced by Dasgupta *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editor-in-Chief of HSDR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Adviser, Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk